Author's response to reviews

Title: Clinimetric properties of the Turkish translation of modified neck disability index

Authors:

Nur Kesiktas (nur.kesiktas@gmail.com)
Emel Ozcan (eozcan@gmail.com)
Howard Vernon (hvernon@cmcc.ca)

Version: 3 Date: 16 October 2011

Author's response to reviews: see over
To the Editor:

We thank the reviewers for their interest in our manuscript and for their helpful recommendations.

For both reviewers: we have revised much of the text for English grammatical and typographic errors. We hope this is to their satisfaction. We have submitted a revised manuscript with “track changes” showing for your review.

For both reviewers: we have added a factorial analysis and have amended the Abstract accordingly.

For reviewer #1: we have added the work of Aslan et al. and have referred to it extensively in the Discussion.

For reviewer #2:

a. there was some confusion regarding the term “randomization”. In fact, the sample was randomly selected; however, there was no ‘randomization’ (i.e. allocation).

b. On Table 2, the lack of a confidence interval for the item ‘driving’ is due to the fact that the ICC was 1.

c. We feel that a statistical analysis is not warranted. This was not recommended by Reviewer #1, and our statistical analyses were performed by a statistician.